Author's response to reviews

Title: Validation of the GenoType(R) MTBDRplus assay for detection of MDR-TB in a public health laboratory in Thailand

Authors:

Rapeepun Anek-vorapong (rapeepan.anek@gmail.com)
Chalinthorn Sinthuwattanawibool (chalinthornS@th.cdc.gov)
Laura Jean Podewils (lpp8@cdc.gov)
Kimberly McCarthy (dxu8@cdc.gov)
Keerataya Ngamlert (supatcharan@yahoo.com)
Busakorn Promsarin (busa_kornp@hotmail.com)
Jay K Varma (jcv9@cdc.gov)

Version: 3 Date: 28 April 2010

To Whom it May Concern:

Please find attached the revised version of our manuscript entitled, “Validation of the GenoType MTBDRplus assay for detection of MDR-TB in a public health laboratory in Thailand” for publication in BMC Infectious Diseases.

We have made the following formatting revisions requested by the editorial staff:

Tables: We have revised all tables to fit onto an A4 page, using portrait orientation.

Abstract: We confirmed with editorial staff that the abstract was already formatted using the correct structure.

Running header: Editorial staff requested this be removed from all pages; we did not have a running header on any of the pages of the manuscript.

Title page: We removed Short Title, Word Count, Abstract wordcount, figure and table counts and Keywords.

Email addresses: We have included email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
Underlining: We have removed underlining as requested.

References: Reference 20 was updated with the current publication information.

Thank you for your review and acceptance of this article. Please do not hesitate to contact me if you have any additional questions or concerns.

Sincerely,

Laura Jean Podewils, PhD, MS
Epidemiologist
International Research and Programs Branch
Division of Tuberculosis Elimination
Centers for Disease Control and Prevention
1600 Clifton Road NE, MS E-10
Atlanta, GA 30333
lpp8@cdc.gov
Tel: (404) 639-5346
Fax: (404) 639-1566